

## **Author's response to reviews**

**Title:** A study of HIV/AIDS related knowledge, attitude and behaviors among female sex workers in Shanghai China

### **Authors:**

Yong Cai ([caiyong202028@163.com](mailto:caiyong202028@163.com))  
Rong Shi ([shirong61@yahoo.com.cn](mailto:shirong61@yahoo.com.cn))  
Tian Shen ([pedy@yahoo.cn](mailto:pedy@yahoo.cn))  
Pei Bei ([babystar009@yahoo.com.cn](mailto:babystar009@yahoo.com.cn))  
Xueqin Jiang ([jiangxueqin@sjtu.edu.cn](mailto:jiangxueqin@sjtu.edu.cn))  
Xiuxia Ye ([xiuxiaye@gmail.com](mailto:xiuxiaye@gmail.com))  
Gang Xu ([smile567@sohu.com](mailto:smile567@sohu.com))  
Shenhui Li ([lsh9907@163.com](mailto:lsh9907@163.com))  
Hong Huang ([huanghong096@hotmail.com](mailto:huanghong096@hotmail.com))  
Meili Shang ([lilyshang1984@163.com](mailto:lilyshang1984@163.com))

**Version:** 8 **Date:** 27 June 2010

**Author's response to reviews:** see over

Dear Prof. Nina Titmus, the Scientific Editor:

Thank you for your reminding of the formatting changes of our manuscript to be made.

Point to Point Reply To the changes

1. Author list: We have changed the order of the list of authors in the submission system to match the manuscript, both in style and order.
2. Abstract: We have changed the Abstract in the submission system to match the manuscript.

Thank you very much.

Please contact us if you have any questions.

We will pay for the article processing charge in 2 weeks.

Best regards,

Hong Huang